Leap or lag: left atrial appendage closure and guidelines

A. John Camm 💿 *

Cardiology Clinical Academic Group, Molecular & Clinical Sciences Institute, St. George's University of London, Cranmer Terrace, London SW17 0RE, UK

Received 19 January 2023; accepted after revision 28 February 2023

Abstract

Atrial fibrillation (AF) is associated with life-threatening thromboembolism. Most emboli stem from thrombosis in the left atrial appendage (LAA). The current treatment of choice is oral anticoagulants (OACs), but a small proportion of patients cannot take OACs predominantly because of the so-called unacceptable bleeding risks. However, many who initially accept OACs subsequently stop therapy or reduce the OAC treatment to a potentially non-effective dose leaving them exposed to thromboembolic risk.

A relatively simple alternative therapy involves the catheter-based insertion of a LAA closure (LAAC) device to prevent thromboembolism from the LAA. There is a considerable evidence base for this therapy consisting of clinical trials and observational data which suggests comparable therapeutic efficacy with a possible small excess of ischaemic strokes.

Although LAAC has been very closely examined by regulators and approved for market release, guidelines from most professional societies give only weak recommendations for use of this device which may be the only known effective therapy available to some at-risk AF patients. Guidance materials from the same societies more enthusiastically endorse LAAC.

Clinical practice is running well ahead of the guidelines because equipoise has been lost by physicians faced with patients for whom they have no other effective therapy. Guideline writers are correct in providing recommendations which are less strong for LAAC than for OACs, for those who are able and willing to take OAC treatment, but for those who are not, a stronger recommendation is needed. But, should the guidelines lag behind or leap ahead of the available evidence?

* Corresponding author. Tel: +44 2087253414; fax + 44 2087253416. *E-mail address*: jcamm@sgul.ac.uk © The Author(s) 2023. Published by Oxford University Press on behalf of the European Society of Cardiology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial License (https://creativecommons.org/licenses/by-nc/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited. For commercial re-use, please contact journals.permissions@oup.com



What's new?

- Medical Practice guidelines and clinical equipoise may be out-of-step with each other. This is the case with the recommendations for LAAC device implantation.
- Therapy approved by regulatory agencies after the submission of some randomised clinical trial and much observational study data is only weakly recommended by professional societies, for an indication for which there is no other treatment which can prevent or reduce life altering complications.
- In such circumstances, guideline writers should consider making a strong recommendation by expert consensus.

Medical guidelines provide great help to the practising clinician, although most clinical decisions for individual patients are not based only on class I guideline recommendations but more often on 'common sense'. Guidelines are difficult to construct and must be prepared by an organization which can ensure a credible writing process and branded endorsement. They have many advantages but also some drawbacks, as summarized and discussed by Rapezzi *et al.*¹ Because the production of guidelines is such a complex and time-consuming process, it is difficult for them to remain up-to-date. The guideline task forces assembled by the European Society of Cardiology (ESC) realistically require a 2-year period for a guideline to be researched, written, reviewed, revised, and published, and the society can at best offer a 4-year cycle between consecutive guidelines on any particular topic. In cardiology arenas which are heavily researched, such as atrial fibrillation, the cycle of 4 years may be too long. One method of dealing with this is to

arrange for updates to particular parts of a guideline which have become outdated, and this approach, which is less onerous, has been used occasionally for international atrial fibrillation guidelines such as those produced by the ESC, the American Heart Association, and the American College of Cardiology.^{2,3} Unless this is done, the guidelines may rapidly become out-of-date and out-of-step with clinical practice as clinical equipoise is lost.

The iterative process of guideline release by various societies, which is rarely synchronized, does not ameliorate but often worsens the situation because of inconsistencies between guidelines. Altogether, this can result in confused and disorganized clinical practice which is not guideline consistent and this may present peril to both the patients and their doctors. Loss of confidence with older recommendations, often superseded by new recommendations from another body, may not just result from the publication of new randomized clinical trial results but also from good-quality although potentially less reliable sources such as non-randomized but carefully collected and well-adjusted observational data. These considerations apply to current guideline recommendations for left atrial appendage (LAA) occlusion.

Left atrial appendage closure

Left atrial appendage closure (LAAC) with a plug-like device was introduced using a catheter-based interventional technique in the early 2000s.⁴ Since then, three relatively large randomized trials, two controlled against dose-adjusted warfarin and one against direct oral anticoagulants (DOACs), together with additional long-term follow-up results have been published.^{5–8} Several competent meta-analyses have been undertaken.^{9–11} Left atrial appendage closure treatment has generally compared well with oral anticoagulant (OAC), both



Figure 1 Search on PubMed for publications relating to 'left atrial appendage closure OR left atrial appendage occlusion' from 2001 to 2022. Reviews and metanalyses and RCT (randomized controlled trial) data are obtained using the appropriate NCBI filters and the observational data are derived by adding 'AND observational OR registry' to the search terms.

with warfarin and with DOAC therapy. There appears to be possibly a small excess of ischaemic strokes when using the LAAC device which is offset by a substantial reduction in non-procedure-related bleeding and mortality, and consequently, treatment with the device may result in net clinical benefit.¹²

With all this available evidence, should those responsible for producing evidence-based guidelines provide a strong recommendation for the use of LAAC? There are at least three major concerns that deter such an outcome:

- (1) The results of the three randomized trials are not entirely consistent and meta-analyses have not unequivocally shown that stroke prevention with LAAC therapy is as effective as with OAC. Such a result might be expected if not all atrial thrombosis arose in the LAA and if the underlying atrial cardiomyopathy, rather than atrial fibrillation itself, was in part responsible for thrombogenesis.¹³
- (2) Only 35% of the oral anticoagulants (OACs) compared with LAAC in the three clinical trials was DOAC therapy, which is now the preferred therapy indicated for stroke prevention in patients with at-risk atrial fibrillation because of their improved safety and efficacy when compared with warfarin in four major pre-approval, controlled randomized trials against dose-adjusted warfarin.^{14–17}
- (3) No randomized controlled clinical trial has yet reported on LAAC placement vs. standard of care, essentially no treatment, in patients unable to take OAC, despite attempts to conduct such trials. For many investigators, equipoise was lost long ago when the results of LAAC used in other settings suggested that this therapy was at least close to being as effective as OAC for stroke prevention but without substantial non-procedure-related bleeding risks.

Although these three issues present a guideline writing dilemma, there are, in addition to the trial data, a host of registries, established by companies, professional societies, and individual investigators, that have reported and continue to report the clinical value of LAAC therapy for a variety of indications, ^{18–22} importantly including patients for whom there is no other safe thromboprophylactic alternative.^{23,24} This

particular group of patients were, of course, excluded in the OAC vs. LAAC clinical trials. The observational data have also allowed the assessment of LAAC treatment against treatment with DOAC therapy.²⁵ Network metanalysis of observational and trial data suggests that LAAC may be marginally less effective than DOAC therapy at preventing ischaemic stroke but highly effective at reducing major and life-threatening bleeding, an advantage that continues for the whole duration of treatment, suggesting that, as time passes post-implantation, this may become an increasingly important benefit of LAAC when compared to lifelong DOAC therapy.^{26,27} The medical literature related to LAAC has burgeoned considerably over the last decade (*Figure 1*).

There are now large-scale ongoing trials comparing LAAC therapy with direct OACs. Other trials are specifically enrolling patients for whom conventional anticoagulation is contraindicated or difficult, such as those with previous intracerebral haemorrhage, advanced chronic kidney disease, or patients for whom previous treatment with anticoagulation has failed to offer protection against ischaemic stroke, etc. Although some of these studies may report in the next 2–3 years, others will take far longer. When these trials are eventually reported, the evidence base for LAAC therapy will be largely complete. Surgical excision or exclusion of the LAA in patients undergoing valve surgery or coronary revascularization has also been undertaken for many years to reduce the likelihood of stroke. Meta-analysis of studies of this therapy shows a significant reduction of stroke risk following surgery.²⁸ However, the recent LAOOS III trial which evaluated LAA surgical exclusion in a large cohort of patients, the majority of whom were also anticoagulated, showed a similar post-operative stroke risk reduction, implying that the elimination of AF-related stroke might eventually be best achieved with a hybrid approach.²

There are clearly two groups of patients for whom LAAC may be a relevant treatment: those for whom OAC is an option and those for whom no currently approved therapy is available. Among those patients who may seem suitable for OAC, there are some who refuse treatment (medication averse) with an OAC^{30,31} and many who fail to



Figure 2 Therapy and guideline development for AF-related stroke thromboprophylaxis. Timelines for the clinical/clinical trial, introduction of anticoagulants, major LAAC therapy trials, and major clinical guidelines and approvals for LAAC therapy. VKA, vitamin K antagonist, LMWH, low-molecular weight heparin, DOACs, direct oral anticoagulants, FXIi, factor XI inhibitors, ESC, first European Society of Cardiology atrial fibrillation guidelines, the numerals following ESC refer to the 2012, 2016 and 2020 guidelines, FDA, Food and Drug Administration, ACCP, American College of Chest Physicians atrial fibrillation guidelines, AHA, American Heart Association/American College of Cardiology/Heart Rhythm Society atrial fibrillation guidelines, CE, conformité européenne (European Conformity). Modified from references.^{41,42}

adhere to or persist with OAC therapy, including DOAC treatment,³² even after a previous ischaemic stroke attributable to atrial fibrillation³³ and despite the best efforts of their physicians. Others include patients unsuitable for anticoagulation only because they are already taking therapy, such as nonsteroidal anti-inflammatory drugs, dual antiplatelet therapy, or powerful *P*-glycoprotein inhibitors, which interact with OAC and potentiate bleeding complications. Essentially, these patients do not receive effective thromboprophylaxis.

Publications stemming from the early days of LAAC rightly emphasized that LAAC was often complicated by serious adverse events.³⁴ However, recent reports point out that LAAC device placement is now a relatively low-risk procedure.^{35–38} Modern, redesigned devices are relatively easy to implant in most patients. The main complications of pericardial effusion, non-occlusion with residual gaps between the device and the atrium, and device-related thrombus and related strokes are now very uncommon.³⁹ In good and experienced hands, LAAC implantation now has a complication rate which is not significantly different to coronary stenting. It is critically important that both institutions and individual operators have adequate experience of device implantation in order to ensure the safety of the technique.⁴⁰ The improved safety of LAAC insertion must improve the benefit–risk of this therapy and lead to a further overall reassessment in comparison to OAC.

The use of the LAAC is becoming relatively commonplace in clinical practice in the USA, but elsewhere, its use is limited. It is estimated that around 200 000 LAAC implants are performed each year in the USA compared with around 16 000 in Europe although the AF population of the USA is roughly half of that in Europe. Almost everywhere, the take-up of this therapy has been patchy and largely in the hands of enthusiasts and pioneers. Therefore, the availability of this treatment is 'postcode-dependent' even for those for whom no other therapy is available. In some regions, the therapy is only available if the patient can join a clinical trial. This unsatisfactory situation is, at least in part, because of the lukewarm recommendations for LAAC use in guidelines from professional societies.

Guidelines

Atrial fibrillation guidelines for the application of LAAC treatment have been offered by the ESC since 2012^2 amid the ongoing development of OACs and LAAC devices (Figure 2).41,42 Many other professional societies have since published their own guidelines, and the ESC guideline was last revised in 2020^{43} (*Table 1*).^{3,44–47} All are similar in trying to identify high-stroke risk atrial fibrillation patients with sufficient bleeding risks from OAC to warrant using an LAAC device. However, their wording is subtly different, for example, OAC use should be 'absolutely contraindicated', 'clear contraindication', or simply 'contraindicated'; the risk of stroke should be 'very high', 'high', 'increased', or not specified; the quality or level of evidence (LOE) may be 'weak', 'LOE: B', or not graded; and the strength or class of recommendation (COR) goes from 'weak' to 'COR: IIb' to 'strong'. There is also a recommendation from the ESC for the use of the LAAC device 'for patients with atrial fibrillation undergoing PCI if a high stroke risk and contraindication for long-term combined antiplatelet plus OAC therapy is present (COR: IIb, LOE: B)'.⁴⁸ Of course, the precise wording of the recommendations and their classifications must be seen as a whole, but suffice it to say that there is a high level of similarity but some inconsistency.

There are other documents which also provide a range of indications for LAAC device implantation. The original FDA regulatory approval for the Watchman device (Boston Scientific) in March 2015 was: 'to reduce the risk of thromboembolism from the LAA in patients with nonvalvular atrial fibrillation who are: at increased risk for stroke and systemic embolism based on CHADS2 [cardiac failure, hypertension, age, diabetes, stroke (doubled)] or CHA2DS2-VASc [congestive heart failure, hypertension, age \geq 75 (doubled), diabetes, stroke (doubled), vascular disease, age 65–74, and sex (female)] scores, are deemed by their physicians to be suitable for warfarin and have an appropriate rationale to seek a non-pharmacologic alternative to warfarin, taking into account the safety and effectiveness of the device compared to warfarin.⁴⁹ A range of devices including the Watchman FLX (Boston Scientific) and the Amplatzer Amulet (Abbott) is now also approved.

Table 1 Recent guidelines

	Recommendation
American College of Chest Physicians Lip et al. ⁴⁴	In patients with AF at high risk of ischaemic stroke who have absolute contraindications for OAC, we suggest using LAA occlusion (weak recommendation,
Asian Pacific Heart Rhythm Society Chao et al. ⁴⁵	low quality evidence) LAA occlusion may be considered for stroke prevention in patients with AF and clear contraindications for long-term anticoagulant treatment (e.g. intracranial bleeding without a reversible cause)
Cardiac Society of Australia and New Zealand Brieger <i>et al.</i> ⁴⁶	LAA occlusion may be considered for stroke prevention in patients with NVAF at moderate to high risk of stroke and with contraindications to oral anticoagulation therapy. GRADE quality of evidence, low ; GRADE strength of recommendation,
American Heart Association, American College of Cardiology and Heart Rhythm Society January <i>et al.</i> ³ Canadian Cardiovascular Society	strong Percutaneous LAA occlusion may be considered in patients with AF at increased risk of stroke who have contraindications to long-term anticoagulation. COR: IIb LOE: B-NR We suggest that percutaneous LAAO be considered for stroke prevention
Andrade et al."' European Society of Cardiology Hindricks et al. ⁴³	in patients with NVAF who are at moderate to high risk of stroke and have absolute contraindications to OAC (weak recommendation; low -quality evidence) LAA occlusion may be considered for stroke prevention in patients with AF and contraindications for long-term anticoagulant treatment (e.g. intracranial bleeding without a

Some current guidelines recommendations illustrating many similarities but some inconsistencies (note bold text).

COR, class of recommendation; LOE, level of evidence.

The form of alternative anticoagulation is no longer restricted to warfarin. Some contraindications for device use are also specified. 50

Several professional society manuscripts which have been published as white papers or consensus documents expand on the detail available in major cardiology society guidelines.^{51,52} The writers generally adhere to the principle that when an OAC can be used, it should take precedence over an LAAC implantation. However, relevant bleeding risks have been

A. Patient not eligible for long- relative contraindications to (-term OAC therapy (absolute or OAC)
High risk for bleeding	Intracranial bleedingGI bleeding
History of major or minor bleeding (±OAC therapy)	 Symptomatic bleeding in critical organ (i.e. ocular, pericardial, and spinal cord) Recurrent epistaxis needing medical attention
Increased risk for bleeding due to physical condition and/or co-morbidities	 Recurrent falls with head trauma and significant musculoskeletal injury Need for additional dual antiplatelet therapy for CAD and stenting Diffuse intracranial amyloid angiopathy Bowel angiodysplasia Severe renal insufficiency/ haemodialysis Blood cell dyscrasia
Inability to take OACs for reasons other than high risk for bleeding	 Intolerance Documented poor adherence to medication Documented variability in the international normalized ratio on warfarin Higher-risk occupation with increased injury potential Patient's choice
B. Thromboembolic event/do	cumented presence of
thrombus in the LAA despi	te adequate OAC therapy
	 Embolic stroke or other systemic thromboembolism on adequate OAC therapy with evidence for

 Table 2
 Indication for the LAAC according to the Munich

Emblie stroke of other systemic thromboembolism on adequate OAC therapy with evidence for thrombus origin from the LAA (malignant LAA)
 Documented thrombus formation in the LAA on adequate OAC therapy

Munich Consensus indications for left atrial appendage closure. OAC, oral anticoagulation; LAA, left atrial appendage (reproduced from reference⁵³).

further explained and the contraindications to OAC use have become more inclusive, not just bleeding risks but also the failure of adequate OAC therapy to prevent ischaemic stroke and the unwillingness or inability of the patient to take medications appropriately. An example, the LAAC indications taken from such a source, the 'Munich Consensus', can be seen in *Table 2.⁵³* In this and other papers, an indication labelled as 'patient choice' is included. Whilst this concept is always important, it presently behoves the physician to carefully explain that the



Figure 3 Recommendations for LAAC therapy from guidelines and guidance documents. Both sources are similar but the guidance documents provide more detail. CHA2DS2-VASc = congestive heart failure, hypertension, age \geq 75 (doubled), diabetes, stroke (doubled), vascular disease, age 65–74, and sex (female); OAC, oral anticoagulation; LAAC, left atrial appendage closure; RF, risk factors.

encouraging evidence that we have so far does not necessarily extend to patients with little or no risk from anticoagulation. The present advice from the European Heart Rhythm Association fully explains this position. 54

Guidelines can be interpreted in a very narrow manner, but a plethora of guidance documents have carefully laid out a richer interpretation of the guidelines by explaining the extensive range of bleeding risks (*Table 2*) and extending the concept of patients unsuitable for anticoagulation to include those who cannot take medication reliably or who have sustained an ischaemic stroke despite well evaluated adequate anticoagulation (*Figure 3*).

Finally, numerous local documents relating to national regulations, payers' conditions, hospital policies, and health technology assessments complete the pantheon of material that must be considered by the physician when contemplating resorting to LAAC implantation.⁵⁵ Much of this advice is related to the initial cost of device implantation, irrespective of potential long-term savings,⁵⁶ and the absence of any strong formal recommendation for the use of the LAAC device from an appropriate professional society such as the ESC.

Resolution

With such a range of major society guidelines, guidance from specialist societies, and national/local regulations, it is easy in the confusion 'not to see the wood for the trees'. What is most important is to encourage and support high-quality research directed towards filling unmet gaps in the evidence base, by encouraging potential device candidates to join randomized trials for which they are eligible. When these trials have been completed, guidelines can be written with full confidence. However, in the meantime, we have patients for whom no guideline applies but the evidence base is not barren. We do have the results from observational studies based on carefully collected and well-adjusted data, together with incomplete but important randomized clinical trial results, that enable us to give cogent advice to our patients. Most conclude that, but for local barriers such as device non-availability, those individuals with the potential to enjoy life despite having atrial fibrillation at high risk of thromboembolic stroke and being unsuitable, for a wide variety of reasons, for guideline-mandated OAC therapy, should be considered for the insertion of an LAAC device. There is, therefore, tension between physicians and the LAAC guidelines and healthcare payers that usually follow guideline recommendations rather than guidance advice (*Figure 4*).

The current ESC recommendation for LAAC (2020) has not changed substantially since it was first included in the 2012 guidelines. It states that 'LAA occlusion may be considered for stroke prevention in patients with AF and contraindications for long-term anticoagulant treatment' and finishes with an example of a contraindication for anticoagulation: 'e.g. intracranial bleeding without a reversible cause'. What should be the class of recommendation for LAAC in a patient like this, for whom there is no alternative thromboprophylaxis? Whilst admitting a shortage of evidence, it may be argued that experts might reach a consensus in favour of a strong class of recommendation, as given by the Australian and New Zealand guideline writers. Other recent ESC guidelines, for example, the 2022 cardio-oncology guidelines, have issued many class I recommendations with only the support of expert



Figure 4 Multiple sources of information are available to the writers of guidelines and guidance material. Guideline writers, who are usually not experts in the specific topic about which they write, pay great attention to randomized controlled trials, systematic reviews, and meta-analyses of this very high-quality data and have relatively little regard for observational data. On the other hand, guidance writers, often experts/enthusiasts, give attention to both types of data. Guidelines strongly influence the development of national/local policies and healthcare-payer agreements whilst physicians are also strongly influenced by guidance material. When there is a conflict between these sources of advice, as it is with left atrial appendage closure (LAAC), the result may be loss of equipoise, clinical confusion, and tension between healthcare professionals and their managers. The patient may be the loser.

The balance between safety and efficacy and timely availability

Regulation of medical devices by the US Food and Drug Administration (FDA) is a *balance between ensuring safety and efficacy and the timely availability of novel* (*therapies*). The 21st Century Cures Act of 2016 shifted this balance toward expediting approval. The Cures Act stated that **the FDA must "consider the role of post-market information in determining the least burdensome means" for approval** can reduce pre-market data on effectiveness "through reliance on post-market controls *Hidano D, et al. JAMA Intern Med.* 2022;183:556-558

... and the same should be said about guidelines to encourage the appropriate use of drugs or devices and allow post-market studies to flourish and furnish further data

Figure 5 Observation on the 21st Century Cures Act of 2016.⁵⁸

consensus,⁵⁷ and in the 2021 ESC AF guidelines, DOAC resumption after ICH was elevated from a class IIb to a class IIa recommendation based on only weak evidence and no randomized controlled trial data.⁴⁴ Without a strong recommendation, healthcare systems can dodge their responsibility and essentially offer no therapy to these patients whose care cannot wait for many years for the completion of more clinical trials. However, when considering any recommendation for LAAC as an alternative for OAC, for example, when a patient prefers not to take oral medication, although the evidence available is considerable, it is insufficient to justify a strong class of recommendation.

There are unacceptable delays involved in fully investigating a medical device before it can be approved by regulators for market release. The need for an expedited approach has been fully recognized in the 21st Century Cures Act in the USA (*Figure 5*) because of the long delays

and huge costs involved in bringing novel medical device therapy to market.⁵⁸ However, this has not been entirely successful because manufacturers have not been sufficiently active in keeping their side of the bargain—post-approval studies. However, this has not been the case with the LAAC device because numerous clinical trials and observational databases have been initiated by the industry or others and are now ongoing. Regulators have changed their approach so as not to needlessly deprive patients of medical device innovations. In order to support the regulatory changes, a somewhat similar approach may be needed for medical guidelines when considering therapies for patients at high risk of life-changing or life-threatening consequences which can be averted at least in part by a well-documented therapy still awaiting full confirmation from long-duration controlled randomized trials. Cautious and comprehensive monitoring of such treatment must then be implemented in order to halt the runaway application of a therapy which is unsafe.

It is not the function of guidelines to impede the use of therapies which have been deemed worthy of market release because of a sufficient but not yet complete evaluation unless subsequent evidence shows the therapy to be inferior or outperformed by other treatments. In the case of the LAAC, its non-inferiority to vitamin K antagonists is broadly accepted, but when compared against DOACS, a relatively new therapy introduced after the critical phase III studies which led to the approval of the first device, the results are similar but less clear. Therefore, any guideline recommendation for patients able to take an anticoagulant can only be relatively restrained and this may persist for some time with the ongoing development of even newer and possibly safer factor XI/XIa inhibitors⁴² although it may by then be the responsibility of the pharma companies to demonstrate that their new drugs outperform the LAAC. However, when no OAC therapy can be effectively prescribed for high-risk patients at risk of stroke, it can be argued that LAAC therapy should qualify now, by expert consensus, for a strong recommendation.

Conclusion

There is an old IBM corporate mantra 'leap before you lag'. The commercial relevance of this is obvious but it should also be considered when guidelines are written. Guideline task forces are conservative and are usually strongly encouraged not to base their recommendations on other than class I evidence. But, because it may take so long to satisfy every requirement to achieve a high level of evidence, valuable therapies, such as LAAC, may become stranded. In consequence, clinical guidelines may lag well behind clinical practice and it is sometimes necessary, especially when patients are left with little or no otherwise effective therapy for their life-threatening illness for guidelines to leap cautiously ahead.

Acknowledgements

Several colleagues have reviewed this manuscript on my behalf.

Conflicts of interest: A.J.C. has received institutional grants and personal fees from Bayer, Boehringer Ingelheim, Bristol Myers Squibb/Pfizer Alliance, Daiichi Sankyo, Menarini, Abbott, and Boston Scientific.

References

- Rapezzi C, Sinagra G, Merlo M, Ferrari R. The impossible interviews-Sherlock Holmes interviews David Sackett: 'how much can we trust the guidelines?' *Eur Heart J* 2021; 42:3422–4.
- Camm AJ, Lip GY, De Caterina R, Savelieva I, Atar D, Hohnloser SH et al. 2012 Focused update of the ESC Guidelines for the management of atrial fibrillation: an update of the 2010 ESC Guidelines for the management of atrial fibrillation. Developed with the special contribution of the European Heart Rhythm Association. Eur Heart J 2012;33: 2719–47.

- January CT, Wann LS, Calkins H, Chen LY, Cigarroa JE, Cleveland JC Jr et al. 2019 AHA/ ACC/HRS focused update of the 2014 AHA/ACC/HRS guideline for the management of patients with atrial fibrillation: a report of the American College of Cardiology/ American Heart Association task force on clinical practice guidelines and the Heart Rhythm Society. J Am Coll Cardiol 2019;74:104–32.
- Nakai T, Lesh MD, Gerstenfeld EP, Virmani R, Jones R, Lee RJ. Percutaneous left atrial appendage occlusion (PLAATO) for preventing cardioembolism: first experience in canine model. *Circulation* 2002;**105**:2217–22.
- Reddy VY, Doshi SK, Sievert H, Buchbinder M, Neuzil P, Huber K et al. Percutaneous left atrial appendage closure for stroke prophylaxis in patients with atrial fibrillation:
 2.3-year follow-up of the PROTECT AF (Watchman Left Atrial Appendage System for Embolic Protection in Patients with Atrial Fibrillation) Trial. *Circulation* 2013;**127**: 720–9.
- Holmes DR Jr, Kar S, Price MJ, Whisenant B, Sievert H, Doshi SK et al. Prospective randomized evaluation of the Watchman Left Atrial Appendage Closure device in patients with atrial fibrillation versus long-term warfarin therapy: the PREVAIL trial. J Am Coll Cardiol 2014;64:1–12.
- Osmancik P, Herman D, Neuzil P, Hala P, Taborsky M, Kala P et al. Left atrial appendage closure versus direct oral anticoagulants in high-risk patients with atrial fibrillation. J Am Coll Cardiol 2020;75:3122–35.
- Osmancik P, Herman D, Neuzil P, Hala P, Taborsky M, Kala P et al. 4-Year Outcomes after left atrial appendage closure versus nonwarfarin oral anticoagulation for atrial fibrillation. J Am Coll Cardiol 2022;79:1–14.
- Reddy VY, Doshi SK, Kar S, Gibson DN, Price MJ, Huber K et al. 5-Year Outcomes after left atrial appendage closure: from the PREVAIL and PROTECT AF Trials. J Am Coll Cardiol 2017;70:2964–75.
- Panikker S, Lord J, Jarman JW, Armstrong S, Jones DG, Haldar S et al. Outcomes and costs of left atrial appendage closure from randomized controlled trial and real-world experience relative to oral anticoagulation. Eur Heart J 2016;37:3470–82.
- Turagam MK, Osmancik P, Neuzil P, Dukkipati SR, Reddy VY. Left atrial appendage closure versus oral anticoagulants in atrial fibrillation: a meta-analysis of randomized trials. J Am Coll Cardiol 2020;76:2795–7.
- Brouwer TF, Whang W, Kuroki K, Halperin JL, Reddy VY. Net clinical benefit of left atrial appendage closure versus warfarin in patients with atrial fibrillation: a pooled analysis of the randomized PROTECT-AF and PREVAIL studies. J Am Heart Assoc 2019;8: e013525.
- Cresti A, Sievert H, Limbruno U, Baratta P, Solari M, De Sensi F et al. Where are clots in atrial fibrillation? Did we have the wrong assumptions over the last decades? *Europace* 2020;**22**:845–6.
- Connolly SJ, Ezekowitz MD, Yusuf S, Eikelboom J, Oldgren J, Parekh A et al. Dabigatran versus warfarin in patients with atrial fibrillation. N Engl J Med 2009;361:1139–51.
- Patel MR, Mahaffey KW, Garg J, Pan G, Singer DE, Hacke W et al. Rivaroxaban versus warfarin in nonvalvular atrial fibrillation. N Engl J Med 2011;365:883–91.
- Granger CB, Alexander JH, McMurray JJ, Lopes RD, Hylek EM, Hanna M et al. Apixaban versus warfarin in patients with atrial fibrillation. N Engl J Med 2011;365:981–92.
- Giugliano RP, Ruff CT, Braunwald E, Murphy SA, Wiviott SD, Halperin JL et al. Edoxaban versus warfarin in patients with atrial fibrillation. N Engl J Med 2013;369:2093–104.
- Hildick-Smith D, Landmesser U, Camm AJ, Diener HC, Paul V, Schmidt B et al. Left atrial appendage occlusion with the Amplatzer[™] amulet[™] device: full results of the prospective global observational study. Eur Heart J 2020;41:2894–901.
- Tzikas A, Shakir S, Gafoor S, Omran H, Berti S, Santoro G et al. Left atrial appendage occlusion for stroke prevention in atrial fibrillation: multicentre experience with the AMPLATZER cardiac plug. *EuroIntervention* 2016;**11**:1170–9.
- Lakkireddy D, Thaler D, Ellis CR, Swarup V, Sondergaard L, Carroll J et al. Amplatzer amulet left atrial appendage occluder versus watchman device for stroke prophylaxis (amulet IDE): a randomized, controlled trial. *Circulation* 2021;**144**:1543–52.
- Boersma LV, Ince H, Kische S, Pokushalov E, Schmitz T, Schmidt B et al. Evaluating realworld clinical outcomes in atrial fibrillation patients receiving the WATCHMAN left atrial appendage closure technology: final 2-year outcome data of the EWOLUTION trial focusing on history of stroke and hemorrhage. *Circ Arrhythm Electrophysiol* 2019;**12**: e006841.
- Cruz-González I, Ince H, Kische S, Schmitz T, Schmidt B, Gori T et al. Left atrial appendage occlusion in patients older than 85 years. Safety and efficacy in the EWOLUTION registry. Rev Esp Cardiol (Engl Ed) 2020;73:21–7.
- Boersma LV, Ince H, Kische S, Pokushalov E, Schmitz T, Schmidt B et al. Efficacy and safety of left atrial appendage closure with WATCHMAN in patients with or without contraindication to oral anticoagulation: 1-year follow-up outcome data of the EWOLUTION trial. *Heart Rhythm* 2017;14:1302–8.
- Tzikas A, Freixa X, Llull L, Gafoor S, Shakir S, Omran H et al. Patients with intracranial bleeding and atrial fibrillation treated with left atrial appendage occlusion: results from the Amplatzer Cardiac Plug registry. Int J Cardiol 2017;236:232–6.
- Nielsen-Kudsk JE, Korsholm K, Damgaard D, Valentin JB, Diener HC, Camm AJ et al. Clinical outcomes associated with left atrial appendage occlusion versus direct oral anticoagulation in atrial fibrillation. JACC Cardiovasc Interv 2021;14:69–78.

- Hanif H, Belley-Cote EP, Alotaibi A, Dvirnik N, Neupane B, Beyene J et al. Left atrial appendage occlusion for stroke prevention in patients with atrial fibrillation: a systematic review and network meta-analysis of randomized controlled trials. J Cardiovasc Surg (Torino) 2018;59:128–39.
- Sahay S, Nombela-Franco L, Rodes-Cabau J, Jimenez-Quevedo P, Salinas P, Biagioni C et al. Efficacy and safety of left atrial appendage closure versus medical treatment in atrial fibrillation: a network meta-analysis from randomised trials. *Heart* 2017;**103**:139–47.
- Tsai YC, Phan K, Munkholm-Larsen S, Tian DH, La Meir M, Yan TD. Surgical left atrial appendage occlusion during cardiac surgery for patients with atrial fibrillation: a meta-analysis. *Eur J Cardiothorac Surg* 2015;**47**:847–54.
- Whitlock RP, Belley-Cote EP, Paparella D, Healey JS, Brady K, Sharma M et al. Left atrial appendage occlusion during cardiac surgery to prevent stroke. N Engl J Med 2021;384: 2081–91.
- Krittayaphong R, Phrommintikul A, Ngamjanyaporn P, Siriwattana K, Kanjanarutjawiwat W, Chantrarat T et al. Rate of anticoagulant use, and factors associated with not prescribing anticoagulant in older Thai adults with non-valvular atrial fibrillation: A multicenter registry. J Geriatr Cardiol 2019;16:242–50.
- Lahaye S, Regpala S, Lacombe S, Sharma M, Gibbens S, Ball D et al. Evaluation of patients' attitudes towards stroke prevention and bleeding risk in atrial fibrillation. Thromb Haemost 2014;111:465–73.
- Beyer-Westendorf J, Ebertz F, Förster K, Gelbricht V, Michalski F, Köhler C et al. Effectiveness and safety of dabigatran therapy in daily-care patients with atrial fibrillation. Results from the Dresden NOAC registry. *Thromb Haemost* 2015;**113**:1247–57.
- Glader EL, Sjölander M, Eriksson M, Lundberg M. Persistent use of secondary preventive drugs declines rapidly during the first 2 years after stroke. Stroke 2010;41:397–401.
- 34. Miller MA, Gangireddy SR, Doshi SK, Aryana A, Koruth JS, Sennhauser S et al. Multicenter study on acute and long-term safety and efficacy of percutaneous left atrial appendage closure using an epicardial suture snaring device. *Heart Rhythm* 2014;**11**: 1853–9.
- Fukuda N, Imamura T, Tanaka S, Kataoka N, Ushijima R, Ueno H et al. Comparison in short-term safety and efficacy between new-generation WATCHMAN FLX and conventional WATCHMAN 2.5 for percutaneous left atrial appendage closure. J Clin Med 2022;11:1618.
- Price MJ, Friedman DJ, Du C, Wang Y, Lin Z, Curtis JP et al. Comparative safety of transcatheter LAAO with the first-generation Watchman and next-generation Watchman FLX devices. *JACC Cardiovasc Interv* 2022;**15**:2115–23.
- Phillips KP, Romanov A, Artemenko S, Folkeringa RJ, Szili-Torok T, Senatore G et al. Combining left atrial appendage closure and catheter ablation for atrial fibrillation: 2-year outcomes from a multinational registry. *Europace* 2020;22:225–31.
- Betts TR, Grygier M, Kudsk JEN, Schmitz T, Sandri M, Casu G et al. Real-world clinical outcomes with a next-generation left atrial appendage closure device: the FLXibility Post-Approval Study. Europace 2023;25:914–21.
- Galea R, Mahmoudi K, Gräni C, Elhadad S, Huber AT, Heg D et al. Watchman FLX vs. Watchman 2.5 in a Dual-Center Left Atrial Appendage Closure Cohort: the WATCH-DUAL study. Europace 2022;24:1441–50.
- Kavinsky CJ, Kusumoto FM, Bavry AA, Bailey SR, Ellenbogen KA, Plet al H. SCAl/ACC/ HRS institutional and operator requirements for left atrial appendage occlusion. J Am Coll Cardiol 2016;67:2295–305.
- Collado FMS, Lama von Buchwald CM, Anderson CK, Madan N, Suradi HS, Huang HD et al. Left atrial appendage occlusion for stroke prevention in nonvalvular atrial fibrillation. J Am Heart Assoc 2021;10:e022274.
- Fredenburgh JC, Weitz JI. New anticoagulants: moving beyond the direct oral anticoagulants. J Thromb Haemost. 2021;19:20–9.
- 43. Hindricks G, Potpara T, Dagres N, Arbelo E, Bax JJ, Blomström-Lundqvist C et al. 2020 ESC guidelines for the diagnosis and management of atrial fibrillation developed in collaboration with the European Association for Cardio-Thoracic Surgery (EACTS): the task force for the diagnosis and management of atrial fibrillation of the European

Society of Cardiology (ESC) developed with the special contribution of the European Heart Rhythm Association (EHRA) of the ESC. *Eur Heart J* 2021;**42**:373–498.

- Lip GYH, Banerjee A, Boriani G, Chiang CE, Fargo R, Freedman B et al. Antithrombotic therapy for atrial fibrillation: CHEST guideline and expert panel report. Chest 2018;**154**: 1121–201.
- Chao TF, Joung B, Takahashi Y, Lim TW, Choi EK, Chan YH et al. 2021 Focused update of the 2017 consensus guidelines of the Asia Pacific Heart Rhythm Society (APHRS) on stroke prevention in atrial fibrillation. J Arrhythm 2021;37:1389–426.
- 46. NHFA CSANZ Atrial Fibrillation Guideline Working Group, Brieger D, Amerena J, Attia J, Bajorek B, Chan KH et al. National Heart Foundation of Australia and the Cardiac Society of Australia and New Zealand: Australian Clinical Guidelines for the Diagnosis and Management of Atrial Fibrillation 2018. *Heart Lung Circ* 2018;27:1209–66.
- Andrade JG, Aguilar M, Atzema C, Bell A, Cairns JA, Cheung CC et al. The 2020 Canadian Cardiovascular Society/Canadian Heart Rhythm Society comprehensive guidelines for the management of atrial fibrillation. Can J Cardiol. 2020;36:1847–948.
- 48. Windecker S, Kolh P, Alfonso F, Collet JP, Cremer J, Falk V et al. 2014 ESC/EACTS guidelines on myocardial revascularization: the task force on myocardial revascularization of the European Society of Cardiology (ESC) and the European Association for Cardio-Thoracic Surgery (EACTS) developed with the special contribution of the European Association of Percutaneous Cardiovascular Interventions (EAPCI). Eur Heart J 2014;35:2541–619.
- FDA Pre-Market approval for the Watchman left atrial appendage closure technology. https://www.accessdata.fda.gov/scripts/cdrh/cfdocs/cfpma/pma.cfm?id=P130013 (20 February 2023, date last accessed).
- 50. FDA WATCHMAN Left Atrial Appendage Closure Device with Delivery System and WATCHMAN FLX Left Atrial Appendage Closure Device with Delivery System -130013/S03. https://www.fda.gov/medical-devices/recently-approved-devices/watchmanleft-atrial-appendage-closure-device-delivery-system-and-watchman-flx-left-atrial-appendage (20 February 2023, date last accessed).
- Meier B, Blaauw Y, Khattab AA, Lewalter T, Sievert H, Tondo C et al. EHRA/EAPCI expert consensus statement on catheter-based left atrial appendage occlusion. *Europace* 2014;**16**:1397–416.
- Casu G, Gulizia MM, Molon G, Mazzone P, Audo A, Casolo G et al. ANMCO/AIAC/ SICI-GISE/SIC/SICCH Consensus Document: percutaneous occlusion of the left atrial appendage in non-valvular atrial fibrillation patients: indications, patient selection, staff skills, organisation, and training. Eur Heart J Suppl 2017;19:D333–53.
- Tzikas A, Holmes DR Jr, Gafoor S, Ruiz CE, Blomström-Lundqvist C, Diener HC et al. Percutaneous left atrial appendage occlusion: the Munich consensus document on definitions, endpoints, and data collection requirements for clinical studies. *Europace*. 2017; 19:4–15.
- Glikson M, Wolff R, Hindricks G, Mandrola J, Camm AJ, Lip GYH et al. EHRA/EAPCI expert consensus statement on catheter-based left atrial appendage occlusion - an update. Europace 2020;22:184.
- 55. Clinical Commissioning Policy: Left Atrial Appendage Occlusion for patients with atrial fibrillation and relative or absolute contraindications to anticoagulation. https://www. england.nhs.uk/wp-content/uploads/2018/07/1692-left-atrial-appendage-occlusion.pdf (20 February 2023, date last accessed).
- Svensson M. Cost-effectiveness analysis of left atrial appendage occlusion in patients with atrial fibrillation and contraindication to oral anticoagulation. *Eur Heart J* 2022; 43:1348–56.
- 57. Lyon AR, López-Fernández T, Couch LS, Asteggiano R, Aznar MC, Bergler-Klein J et al. 2022 ESC Guidelines on cardio-oncology developed in collaboration with the European Hematology Association (EHA), the European Society for Therapeutic Radiology and Oncology (ESTRO) and the International Cardio-Oncology Society (IC-OS). Eur Heart J 2022;43:4229–361.
- Hidano D, Dhruva SS, Redberg RF. US Food and Drug Administration-mandated postmarketing studies for high-risk cardiovascular devices approved 2015-2019. JAMA Intern Med 2022;182:556–8.